Massachusetts Institute of Technology Handout 36 6.857: Network and Computer Security December 4, 2001 Professor Ronald L. Rivest

Survey

Your suggestions can help improve 6.857 for students next year. You may send us responses via interdepartmental mail or email to 6.857-staff@mit.edu.

1. What did you like best about 6.857? What was your favorite lecture topic?

2. What did you like least about 6.857? What was your least favorite lecture topic?

3. What lecture topics would you like to have seen covered that we did not?

4. How can the staff more efficiently help you, assuming that MIT will only hire 1.5 TAs.

5. Comment on the usefulness of the number, location, and time of office hours.

6. Do you plan to study or work in the realm of computer security?

7. Would you like to TA or grade for 6.857 in the future? If so, write your name.

8. What did you think of the problem sets? This year, we had many programming problem sets. Were these useful compared to pure theory? Fun?

9. Did your group communicate well? Why or why not? What advice would you give students next year in selecting and maintaining a group?

2 Handout 36: Survey 10. Did the midterm accurately reflect your understanding of the lecture material?

11. Do you have any constructive suggestions for next year concerning:

• Lectures

• Homework

• TAs

• Midterm

• Term projects

(fold here)

Prof. Ron Rivest NE43-324